Author’s response to reviews

Title: Impact of initial dialysis modality on mortality: a propensity-matched study

Authors:

Bård Waldum-Grevbo (tleivest@ous-hf.no)
Torbjørn Leivestad (tleivest@ous-hf.no)
Anna Reisæter (areisate@ous-hf.no)
Ingrid Os (ingrid.os@medisin.uio.no)

Version: 2 Date: 05 Oct 2015

Author’s response to reviews:

Thank you for accepting the manuscript for publication in BMC nephrology.

1. I have included all author e-mails on the title page
2. A list of abbreviations is added preceding the competing interests section
3. The study was not funded by any grant, therefore the acknowledgements section was not changed

Bård Waldum-Grevbo